

Article



Geometric Control and Structure-at-Infinity Control for Disturbance Rejection and Fault Compensation Regarding Buck Converter-Based LED Driver

Jesse Y. Rumbo-Morales ¹^(D), Jair Gómez-Radilla ¹, Gerardo Ortiz-Torres ¹^(D), Felipe D. J. Sorcia-Vázquez ¹^(D), Hector M. Buenabad-Arias ^{2,3}, Maria A. López-Osorio ⁴, Carlos A. Torres-Cantero ^{5,6}^(D), Moises Ramos-Martinez ¹^(D), Mario A. Juárez ⁷^(D), Manuela Calixto-Rodriguez ², Jorge A. Brizuela-Mendoza ⁸^(D), Jesús E. Valdez-Resendiz ^{9,*}^(D)

- ¹ Computer Science and Engineering Department, University of Guadalajara, Ameca 46600, Mexico; jesse.rumbo@academicos.udg.mx (J.Y.R.-M.); jair.gomez7620@alumnos.udg.mx (J.G.-R.); gerardo.ortiztorres@academicos.udg.mx (G.O.-T.); felipe.sorcia@academicos.udg.mx (F.D.J.S.-V.); moises.ramos@valles.udg.mx (M.R.-M.)
- ² División Académica de Mecánica Industrial, Universidad Tecnológica Emiliano Zapata del Estado de Morelos, Av. Universidad Tecnológica No. 1, Col. Palo Escrito, Emiliano Zapata 62760, Mexico; hastachura abad@ura en una (UMB, A) en encuelo eliste@uta adu enu (MCC, B)
- hector.buenabad@uaem.mx (H.M.B.-A.); manuelacalixto@utez.edu.mx (M.C.-R.)
- ³ Centro de Investigación en Ingenierías y Ciencias Aplicadas de la UAEM, Cuernavaca 62209, Mexico
 ⁴ Natural and Exact Sciences Department, University of Guadalajara, Ameca 46600, Mexico; maria.osorio@academicos.udg.mx
- ⁵ Tecnológico Nacional de México Campus Colima, Av. Tecnológico # 1, Col. Liberación, Villa de Álvarez 28976, Mexico; atorres@colima.tecnm.mx
- ⁶ Facultad de Ingeniería Mecánica y Eléctrica, Universidad de Colima, Carretera Colima-Coquimatlan km 9, Valle de las Huertas, Coquimatlán 28400, Mexico
- ⁷ TecNM/ITS Irapuato, Irapuato 36821, Mexico; mario.jb@irapuato.tecnm.mx
- ⁸ Exact Sciences and Methodologies Department, University of Guadalajara, Ciudad Guzmán 49000, Mexico; jorge.brizuela@cusur.udg.mx
- School of Engineering and Sciences, Tecnologico de Monterrey, Av. Eugenio Garza Sada 2501 Sur, Tecnológico, Monterrey 64700, Mexico
- Correspondence: jesusvaldez@tec.mx

Abstract: Currently, various light-emitting diode (LED) lighting systems are being developed because LEDs are one of the most used lighting sources for work environments, buildings, homes, and public roads in terms of some of their applications. Similarly, they have low energy consumption, quick responses, and excellent optimal performance in their operation. However, these systems still need to precisely regulate lighting, maintain stable voltage and current in the presence of faults and disturbances, and have a wide range of operations in the event of trajectory changes or monitoring tasks regarding the desired voltage and current. This work presents the design and application of two types of robust controllers (structure-at-infinity control and geometric control) applied to an LED driver using a buck converter. The controllers aim to follow the desired trajectories, attenuate disturbances at the power supply input, and compensate for faults in the actuator (MOSFET) to keep the capacitor voltage and inductor current stable. When comparing the results obtained with the two controllers, it was observed that both present excellent performance in the presence of constant disturbances. However, in scenarios in which variable faults and path changes are implemented, the structure-at-infinity control method shows an overimpulse of output voltage and current ranging from 39 to 42 volts and from 0.3 to 0.45 A, with a margin of error of 1%, and it can generate a failure in the LED driver using a buck converter. On the other hand, when using geometric control, the results are satisfactory, achieving attenuating constant disturbances and variable faults, reaching the desired voltage (40 v to 35 v) and current (0.3 to 0.25 A) with a margin of error of 0.05%, guaranteeing a system without overvoltages or the accelerated degradation of the components due to magnetic conductivity.



Citation: Rumbo-Morales, J.Y.; Gómez-Radilla, J.; Ortiz-Torres, G.; Sorcia-Vázquez, F.D.J.; Buenabad-Arias, H.M.; López-Osorio, M.A.; Torres-Cantero, C.A.; Ramos-Martinez, M.; Juárez, M.A.; Calixto-Rodriguez, M.; et al. Geometric Control and Structure-at-Infinity Control for Disturbance Rejection and Fault Compensation Regarding Buck Converter-Based LED Driver. *Mathematics* **2024**, *12*, 1277. https://doi.org/10.3390/ math12091277

Academic Editor: Jiangping Hu

Received: 28 February 2024 Revised: 16 April 2024 Accepted: 19 April 2024 Published: 23 April 2024



Copyright: © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). Keywords: LED driver; geometric control; structure-at-infinity control; disturbances and faults

MSC: 94C12; 93B52

1. Introduction

With the development of LED lighting and the high demands of consumers, LED chip manufacturers are continually improving and innovating. The power LED is low-cost, has a long operating life, and has a high luminous flux at the output. The intensity it presents can be modulated quickly (on the order of MHz), and it can be easily adapted. However, properly handling this class of LEDs requires a driver (DC–DC converter) to reproduce the appropriate waveforms according to the required intensity. Different methods are proposed for this type of application regarding LEDs; some of them use Radio Frequency (RF) amplifiers with necessary characteristics regarding the response speed; these have the advantage of being linear; however, these circuits have certain deficiencies compared to others. Any methodology or scheme can be adapted to power LED technology, which requires a constant voltage; therefore, it is important to maintain the system's stability in the face of disturbances that may exist as unwanted inputs. These disturbances can cause high voltage or voltage peaks, and, as a consequence, the temperature of the diodes that belong to the bulb increases significantly, and there is a fault in a short period, resulting in damage to the power LED.

Therefore, it is necessary to investigate more details regarding the analysis and design of the controllers to reject disturbances and increase the efficiency of the LED driver using a buck converter. The idea of rejecting or attenuating disturbances is to obtain better results in stabilizing the voltage and noise compensation in this LED driver, achieving better performance and constant luminosity regarding the power LED.

Disturbance rejection is an important issue that must be considered in linear and nonlinear systems to improve the stability and mitigate the effects of these unwanted inputs or system faults, which may be known or unknown values. In the literature, some works of certain researchers who have studied this problem stand out, such as [1–3]. Optimal controllers are receiving the attention of research centers since there is promising potential in the performance of non-linear systems. Some controllers show great performance when experimental studies are performed and demonstrate effectiveness [4–6]. Similarly, Fault-Tolerant Controllers have been developed that can keep the system or process stable and compensate for faults in the actuators or sensors [7,8]. Guaranteeing a stable system with fast responses using controllers is satisfactory and can ensure that the process has optimal and efficient production [9-12]. A variety of controllers have been applied to electrical systems and have focused on maintaining stable current or voltage, derived from the elements considered in the circuit, whose objective of the controllers is to follow the desired trajectories, taking into account the dynamic responses of the output variables [13–15]. In some research works, the decomposition into subsystems with fast or slow responses is necessary to design the controller and to be able to attenuate the disturbances in linear or non-linear systems [16–18].

Furthermore, [19] presents a Current Mode Control for a DC–DC buck converter based on the power LED. The objective was to reduce the number of sensors to maintain the performance of the LED driver. The tests carried out were based on numerical analyses to validate the controller.

Subsequently, in the research carried out by [20], a Dual-Frequency Buck Driver (DFBD) was developed that uses *SiC* diodes (silicon carbide diodes) for LED lighting systems. The design of the PID control (Proportional, Integral, and Derivative Control) was based on Small-Signal Modeling (SSM) and achieved a closed loop for the DFBD. The control of the DFBC is performed to achieve and maintain a stable desired output

voltage and current. Experimentally, the regulation of the output voltage in the presence of disturbances is possible.

Additionally, ref. [21] conducted an analysis of the dynamic performance of the stepdown driver developed with a non-linear high-power LED model. A second-order model represents the non-linear model of the LED to define the dynamic behavior. A PI control was designed to test the linear and non-linear models, resulting in a better response using the second-order non-linear model.

Similarly, in [22], an LED driver with a high Power Factor (PF) was researched and developed with a universal AC voltage input. A dual sampling loop feedback control was designed for the LED driver application. The controller was intended to regulate the switch to adapt the AC input voltage and stabilize the disturbance. It was possible to obtain a quick response with the feedback control, avoiding overheating or peaks that would damage the LED. This LED driver included a circuit architecture that is more economical than others in the literature.

In some electrical systems, it is not only necessary to reject disturbances; if the system tends to be unstable, the system must stabilize at a desired reference. Currently, there are automatic control works that only guarantee the stability of the system or processes and very few works that mitigate disturbances and at the same time stabilize the system.

In [23], research on the control of the DC–DC converters applied to LED drivers is presented. The control design was based on a variable inductor, which can control the converter's output to keep the current through the power LED stable. Likewise, with this magnetic control technique, there was an improvement in the control of DC–DC converters with an LED drive and the combination of it with other parameters, such as duty cycle or frequency, which improved the converter's performance.

In [24], a series of steps were proposed to model a Current Mode Controlled (CMC) flyback LED controller with a stabilizing ramp. The outstanding results were experimentally validated with the proposed control, which can be implemented with a compensation circuit.

Furthermore, in [25], a step-down-type LED driver was developed with automatic current exchange and a high step-down voltage conversion ratio but without complex control. The system includes the primary circuit and the feedback loop. The main circuit was derived from an interlaced buck converter with low voltage and a high step-down ratio [26]. The feedback control circuit adopts an integral proportional control (PI) to keep the current stable.

Another study [27] commented that LED activation circuits must have automatic control of the constant output current and/or constant luminance control in the presence of disturbances. The research focused on proposing an LED driver consisting of a step-up-type DC–DC converter with constant output current control and constant luminance control in which a single integrated system can adapt power and current control without an external circuit or system that can be used as an auxiliary when faults or disturbances occur.

Similarly, in [28], an analysis of the different topologies of standard DC–DC converters that can be adapted to supply a constant current to many LEDs was developed. The research shows that some developments regarding efficient LED drivers via pulsating currents may be possible; this simplifies the use of converters and automatic control and reduces the components in the circuit. The analysis shows that, under certain operating conditions, it is not necessary to use sensors and automatic current controllers to stabilize the average current in an LED.

The novelty of this work is presenting two control laws (geometric control and structure-at-infinity control) regarding an LED driver using a buck converter since there are no works related to these control theories in the literature that can attenuate the disturbances at the input and compensate for the faults in the actuator.

The main contribution of this article is designing and implementing two robust controllers on the LED driver using a buck converter, with the main objective of compensating for the effect of the specific faults that occur in the actuator (MOSFET) and attenuating the disturbances in the power supply, which can critically damage the LED driver. This work also provides a theory not used in these circuits that requires control laws to obtain excellent performance, achieving stable values regarding the desired output of the capacitor voltage and inductor current to avoid overvoltage or short circuits, as well as magnetization or accelerated magnetic conductivity, which can generate degradation and poor performance regarding the components of the electrical system.

The remaining sections of the paper have the following organization: Section 2 describes the mathematical model of the LED driver using a buck converter, obtaining a representation of the state space with disturbance and faults. Section 3 examines the theory and design of the structure regarding infinity control and geometric control. Section 4 presents the analytical cases of the power supply disturbance rejection and actuator fault compensation using robust controllers. Finally, Section 5 includes the results and discussion, analyzing the controllers' signals and the responses obtained from the capacitor voltage and inductor current used in the LED driver using a buck converter.

2. Modeling of LED Driver Using a Buck Converter

It is important to mention that this work focused on carrying out simulations with real parameters obtained from the literature to achieve an approximation of the experimental compartment that the LED driver would present with robust controllers that can attenuate disturbances and compensate for faults. Therefore, this section addresses the study case considering an LED driver with a buck converter with a corresponding mathematical model. The LED driver scheme is shown in Figure 1.



Figure 1. LED driver with buck converter.

From a series of algebraic steps (see Appendix A), it is possible to obtain a mathematical model in the state space that defines the dynamic behavior of the electrical system, considering $x_1 = i_L$ and $x_2 = V_C$ as state variables. The model obtained is shown below:

$$\dot{x}_1 = \frac{1}{L} V_{in} - \frac{1}{L} x_2, \tag{1}$$

and

$$\dot{x}_2 = \frac{1}{C_c} x_1 - \frac{1}{R_L C_c} x_2 + \frac{1}{R_L C_c} V_{ls}.$$
(2)

The linear system obtained (Equations (1) and (2)) is expressed as follows in continuoustime state space representation:

$$\dot{x} = Ax + Bu + R,$$

$$y = Cx,$$
(3)

where $A \in \mathbb{R}^{n \times n}$ is a matrix associated with the vector states ($x \in \mathbb{R}^{n \times 1}$), $B \in \mathbb{R}^{n \times m}$ is a matrix associated with the input vector ($u \in \mathbb{R}^{m \times 1}$), and $C \in \mathbb{R}^{n \times r}$ is the matrix associated with the outputs ($y \in \mathbb{R}^{r \times 1}$) of the system, and R is a constant.

Substituting Equations (1) and (2) into Equation (3), it is possible to obtain the following matrix equation:

$$\begin{bmatrix} \dot{x}_1 \\ \dot{x}_2 \end{bmatrix} = \begin{bmatrix} 0 & -\frac{1}{L} \\ \frac{1}{C_c} & -\frac{1}{R_L C_c} \end{bmatrix} \begin{bmatrix} x_1 \\ x_2 \end{bmatrix} + \begin{bmatrix} \frac{V_{in}}{L} \\ 0 \end{bmatrix} u + \begin{bmatrix} 0 \\ \frac{V_{ls}}{R_L C_c} \end{bmatrix}.$$
 (4)

After defining the model of the LED driver using a buck converter represented by Equation (4), we continued to define the disturbances and faults that may occur in this electrical system.

2.1. Inclusion of the Disturbance and Fault in the LED Driver Model

For the system to be able to reject disturbances or compensate for the effect of a fault, it is necessary to design a robust control law; this must be adequate to attenuate and mitigate known disturbances or faults. In the case of the LED driver with a buck converter, we assume that the disturbance or fault of the plant can be estimated; these disturbances and faults can be

Actuator fault (MOSFET):

The actuator regulates the power flow of the plant, and that can alter the voltage and current output of the buck converter. Therefore, the actuator fault is modeled as an external additive signal, denoted k, added to the original control signal u; that is, $u_f = u + k$. This additive signal represents the failure in actuator performance and is quantified by the parameter μ , which is known as the loss of effectiveness (LoE) of the actuator and goes from zero to one; that is, $0 \le \mu \le 1$.

The actuator fault, represented by $k = -\mu u$, means that the control signal u is multiplied by the negative effectiveness loss factor μ to model how the fault affects the actuator. This can be understood as a loss of effectiveness in generating the PWM signal by the actuator.

Faults in the power system's storage components can be considered partial faults of the actuator. Physically, these components (inductors and capacitors) can degrade over time as a result of factors such as aging, magnetization, and magnetic conductivity. These degradations can affect the system's ability to store and release energy efficiently, which is reflected in the control signal u and, therefore, in the signal u_f that includes the fault; that is,

$$u_f = \delta u,$$

$$\delta = (1 - \mu),$$

$$u_f = (1 - \mu)u = u - \mu u;$$

(5)

therefore,

$$u_f = u + k. \tag{6}$$

Applying Equation (6) to Equation (3), the linear system obtained is as follows:

$$\dot{x} = Ax + B(u+k) + R,$$

= $Ax + Bu + Bk + R.$ (7)

Therefore, applying the system represented by Equation (7) that considers the fault system, it is possible to obtain the following matrix equation:

$$\begin{bmatrix} \dot{x}_1 \\ \dot{x}_2 \end{bmatrix} = \begin{bmatrix} 0 & -\frac{1}{L} \\ \frac{1}{C} & -\frac{1}{CR} \end{bmatrix} \begin{bmatrix} x_1 \\ x_2 \end{bmatrix} + \begin{bmatrix} \frac{V_{in}}{L} \\ 0 \end{bmatrix} u + \begin{bmatrix} \frac{V_{in}}{L} \\ 0 \end{bmatrix} k + \begin{bmatrix} 0 \\ \frac{V_{ls}}{CR} \end{bmatrix}.$$
(8)

Disturbance in power supply:

The disturbance will be generated in the power supply and may alter the input voltage. The disturbed system for the LED driver model using a buck converter has the following expression:

$$\dot{x} = Ax + Bu + Ed + R,$$

$$y = Cx.$$
(9)

The alteration in the system is presented using Equation (9), where *E* is the unwanted input matrix and is associated with *d*, represented as a disturbance in the power supply, and *u* represents the variation in the useful cycle; this is shown as an additive signal in the state space [15]:

$$\begin{bmatrix} \dot{x}_1\\ \dot{x}_2 \end{bmatrix} = \begin{bmatrix} 0 & -\frac{1}{L}\\ \frac{1}{C} & -\frac{1}{CR} \end{bmatrix} \begin{bmatrix} x_1\\ x_2 \end{bmatrix} + \begin{bmatrix} \frac{V_{in}}{L}\\ 0 \end{bmatrix} u + \begin{bmatrix} \frac{W}{L}\\ 0 \end{bmatrix} d + \begin{bmatrix} 0\\ \frac{V_{is}}{CR} \end{bmatrix}.$$
 (10)

After defining and structuring the systems (Equations (8) and (10)) with disturbances and faults, we proceed to simulate the electrical system without control and contemplating the disturbances and faults in the actuator.

2.2. Simulation of LED Driver Model Using a Buck Converter

At this point, the electrical system simulation is carried out considering the parameters of the LED driver using a buck converter. The parameters used are shown in Table 1. The simulations presented at this point are in an open loop, with disturbance in the power supply and with faults in the actuator (MOSFET).

Table 1. Power LED driver parameters.

Parameter and Symbol	Value	Unit
Capacitor C _c	0.48	μF
Voltage V _{in}	80	V
Voltage of the LED V_{ls}	32.51	V
Inductor L	5.17	mH
Resistor of the LED model R_L	22.54	Ω
Duty cycle W	0.495	PWM

With an implementation time of 12.5 μ s. Using Equation (3), the matrix equations are as follows:

$$A = \begin{bmatrix} 0 & -193.42\\ 2083000 & -92428.27 \end{bmatrix}, B = \begin{bmatrix} 15473.88\\ 0 \end{bmatrix}, R = \begin{bmatrix} 0\\ 92428.27 \end{bmatrix}, C = \begin{bmatrix} 0 & 1 \end{bmatrix}, C = \begin{bmatrix} 0 &$$

Figure 2 shows the dynamics of the system in an open loop and without disturbances or faults that affect the LED driver. It is observed that the system stabilizes in approximately one-thousandth of a second. The operating points of the open-loop system are $i_L = 0.3145$ A, $V_C = 39.6$ v, with constant input W = 0.495, which represents a PWM of a duty cycle 50%.



Figure 2. (a) Open–loop capacitor voltage output; (b) open–loop inductor current output.

When a fault or disturbance occurs in the system, Equations (7) and (9) are used to simulate the electrical system. Numerically, the matrix equations are as follows:

After incorporating the disturbance or fault using Equations (10) and (8), the matrices with numerical values would be as follows.

• Matrix values with fault:

$$A = \begin{bmatrix} 0 & -193.42\\ 2083000 & -92428.27 \end{bmatrix}, B = \begin{bmatrix} 15473.88\\ 0 \end{bmatrix}, B = \begin{bmatrix} 15473.88\\ 0 \end{bmatrix}, R = \begin{bmatrix} 0\\ 92428.27 \end{bmatrix}, C = \begin{bmatrix} 0 & 1 \end{bmatrix}.$$

where $B \in \mathbb{R}^{n \times m}$ is a matrix associated with the fault values ($k \in \mathbb{R}^{m \times 1}$).

• Matrix values with disturbance:

$$A = \begin{bmatrix} 0 & -193.42\\ 2083000 & -92428.27 \end{bmatrix}, B = \begin{bmatrix} 15473.88\\ 0 \end{bmatrix}, E = \begin{bmatrix} 95.745\\ 0 \end{bmatrix}, R = \begin{bmatrix} 0\\ 92428.27 \end{bmatrix}, C = \begin{bmatrix} 0 & 1 \end{bmatrix}.$$

where $E \in \mathbb{R}^{n \times m}$ is a matrix associated with the perturbation values ($d \in \mathbb{R}^{m \times 1}$).

For the fault tests, a sine wave-type fault actuator (MOSFET) was applied, as shown in Figure 3a. This type of fault is variable, with a magnitude of 0.2 to -0.2. Likewise, this partial fault in the actuator can cause degradation in the inductor or capacitor.



Figure 3. (a) Sinewave-type variable fault; (b) constant step-and ramp-type disturbances.

On the other hand, in the test with perturbation (defined in Figure 3b), it is defined as a double perturbation. The values of this disturbance vary from 0 v to 4 v and from 0 v to 3 v. This is defined as a disturbance that causes excessive voltage drops or increments, which may be temporary or permanent, as well as current injections at startup, harmonic distortions, or overvoltages. These unwanted inputs can cause unusual power consumption.

Figure 4 shows the dynamics of the open-loop system with disturbances and faults that affect the MOSFET or generate an overvoltage in the power supply of the LED driver. It is observed that the system does not achieve stabilization after applying the perturbation and fault, generating oscillations from 50 v to 30 v (disturbance) and 45 v a 30 v when the fault occurs (see Figure 4b). On the other hand, the inductor current varies from 0.65 A to 0 A (disturbance) and from 0.39 A to 0.30 A when the fault occurs (see Figure 4a). These values damage and affect the operation of the LED driver using a buck converter.

From these simulations, possible solutions can be analyzed and defined to mitigate disturbances and compensate for faults in the electrical system. There are several ways to analyze and solve the problem of disturbance rejection or fault compensation (known or unknown). In particular, for this work, the decoupling of disturbances or known faults is developed. Finally, on the basis of these results, the design of the control laws can be addressed to stabilize, attenuate, compensate, and follow the desired trajectories of the capacitor voltage and inductor current. This is described in the next section.



Figure 4. (a) Open-loop capacitor voltage output; (b) open-loop inductor current output.

3. Theory and Design of Robust Controllers

In theory and application, disturbance rejection or fault compensation are important topics in control systems. These problems have been studied in many industrial systems or processes where more disturbances or faults have been found, and, at the same time, more methods are proposed to reject these disturbances or compensate for faults.

Sometimes, not only is it necessary to reject the disturbance or compensate for the fault but, if the system is unstable, control would be necessary to stabilize the system [29], and, in this case, an integral state feedback control method is necessary to make it non-zero and reject perturbations, compensate for the fault, and follow desired trajectories with zero steady-state error.

Therefore, the first theory designed and applied in this work is structure-at-infinity control, and this is presented in the following subsection.

3.1. Structure at the Infinity Control

The state equation can be written as

$$\dot{x} = Ax + Bu, \quad y = Cx. \tag{11}$$

The state feedback control scheme is used as follows:

$$u = Fx. (12)$$

Subsequently, the characteristic equation of the system (A, B, C) is calculated to obtain the roots:

$$Ps = det(sI - A) = 0.$$
(13)

F is defined as follows:

$$F = [f_{11} f_{12} f_{13} \dots f_{1n}]. \tag{14}$$

Substituting this *F* into Equation (11), one obtains the following:

$$\dot{x} = (A + BF)x, \quad y = Cx. \tag{15}$$

Now, the characteristic polynomial of this new system is determined:

$$Pc = det(sI + A + BF) = 0.$$
(16)

Next, the desired locations of the closed-loop poles are chosen. The system is required to have a fast convergence time and reasonable damping.

The characteristic polynomial is set equal to a desired polynomial with roots close to zero.

$$Pc = Pd. (17)$$

Once the matrix F has been determined, the state feedback gain matrix must be calculated [30].

Subsequently, for the disturbance rejection problem, the following linear system (A, B, C, E) is considered, provided by

$$\dot{x} = Ax + Bu + Ed, \quad y = Cx. \tag{18}$$

where $t \ge 0$, $d \in \mathcal{D} \subseteq \mathbb{R}^m$ is the perturbation and *E* is the matrix representation of the linear application $E : \mathcal{D} \to \mathcal{X}$.

The transfer function of the system (A, B, C) is

$$H(s) = C(sI - A)^{-1}B.$$
(19)

The transfer function of the system (A, B, C, E) with perturbation is

$$H^{d}(s) = C(sI - A)^{-1}E.$$
(20)

The transfer functions H(s) and $H^{d}(s)$ are rational and strictly proper [31,32].

Suppose that the state $x(\bullet)$ and the disturbance $d(\bullet)$ are measurable as shown in Figure 5.



Figure 5. Perturbation problem with state feedback control and structure at the infinity control.

Consider a state feedback control function that is provided by

$$u = Fx + Jd. \tag{21}$$

where $F \in \mathbb{R}^{m \times n}$ and $J \in \mathbb{R}^{m \times m}$ are matrices representing the linear maps $F : \mathcal{X} \to \mathcal{U}, J : \mathcal{D} \to \mathcal{U}$.

Replacing u = Fx + Jd in the system (*A*, *B*, *C*, *E*), we obtain the following:

$$\dot{x} = (A + BF)x + (BJ + E)d, \quad y = Cx.$$
 (22)

To determine the transfer function $d \rightarrow y$, the Laplace transform is applied and is obtained:

$$x(s) = (sI - A - BF)^{-1}(BJ + E)d(s), \ y(s) = Cx(s).$$
(23)

Then,

$$y(s) = C(sI - A - BF)^{-1}(BJ + E)d(s).$$
(24)

Therefore, the transfer function is

$$\frac{y(s)}{d(s)} = C(sI - A - BF)^{-1}(BJ + E), \ J(s) = C(sI - A - BF)^{-1}(BJ + E).$$
(25)

After calculating *F*, we find *J* using the implementation of structure-at-infinity control to reject or attenuate the disturbance in the output.

If the infinity structure of the system and the infinity structure of the unstable system with perturbation are the same, then the system can attenuate the perturbations. Therefore, a control law can be implemented to mitigate disturbances in the controlled variable.

Definition 1 ([29]). The disturbance rejection problem has a solution if there exist a state feedback $F : \mathcal{X} \to \mathcal{U}$ and a function $J : \mathcal{D} \to \mathcal{U}$ such that the transfer function of the output and disturbance is null due to the control function u = Fx + Jd.

If the zero at the infinity of the system (A, B, C) and the zero at the infinity of the system with disturbance (A, B, C, E) are equal, the system allows the perturbation to be rejected; that is, a control function that rejects disturbances at the system outputs can be found to attenuate the disturbances [33].

Definition 2. A complex function f(s) is proper if $\lim_{s \to \infty} (s)$ is finite when $|s| \to \infty$.

Definition 3. A complex function f(s) is strictly proper if $\lim_{s \to \infty} f(s)$ is zero when $|s| \to \infty$.

Definition 4. *A rational function is proper when* $gr(num) \le gr(den)$ *and strictly proper when equality is not satisfied.*

Definition 5 ([34]). A rational function $f(s) \neq 0$ has a zero at $s = \infty$ if

$$\mathbf{f}(s) = \frac{1}{s^r} g(s),\tag{26}$$

where

- 1. *r* is a positive number.
- 2. $\lim_{s\to\infty} g(s) < \infty$.
- 3. $\lim_{s\to\infty} g(s) \neq 0.$

This means that f(s) has a zero at the infinity of order r.

Definition 6 ([33]). *The disturbance rejection problem has a solution if the zero-to-infinity ratio of* H(s) *and* $H^{d}(s)$ *is equal to*

$$\sum_{\infty} \left[H(s), H^d(s) \right] = \sum_{\infty} [H(s), 0].$$
(27)

The following propositions are equivalent [33].

- 1. The following equality is satisfied $\sum_{\infty} \left[H(s), H^d(s) \right] = \sum_{\infty} [H(s), 0].$
- 2. $\lim_{s\to\infty} J(s) < \infty$.
- 3. The disturbance is rejected by the feedback u = Fx + Jd, such that $J(s) = C(sI A BF)^{-1}(BJ + E) = 0$.

Therefore, the control function u = Fx + Jd is used, but *F* is obtained with (12) to stabilize the system, and *J* is obtained with (25) to attenuate disturbance (the proof of the theorem can be consulted in the work presented by [29]).

After defining the structure-at-infinity control theory, we continue by defining geometric control theory and design, which is described in the following subsection.

3.2. Geometric Control

The geometric elements are supported by exhaustive computational machinery; it is quite difficult to obtain a complete overview of them; there are already various works in the literature that address this type of control approach [35–38].

The geometric approach is a set of tools, properties, and algorithms that allow us to analyze and define precise terms for specific properties of dynamic systems and to obtain different views of how we can approach the system or process with automatic control. New approaches have been developed; however, they do not have greater robustness against perturbations in new problems of linear and non-linear systems [39–41].

In this article, we present the geometric control approach to reject the disturbance and fault compensation. The fundamental tool is the invariant subspace in which the kernel of a linear transformation, the image of a linear transformation, or the inverse image must be found (see Figure 6) [42].



Figure 6. Representation of a subspace.

Definition 7. For a given matrix C of dimension $p \times n$, ker(C) can be defined in the following way:

$$\ker(C) = \{ x \in \mathcal{X} \mid Cx = 0 \}.$$
(28)

dim ker(*C*) is a subspace of $\mathcal{X} \subseteq \mathbb{R}^n$. This subspace can be defined as follows:

$$\dim \ker(C) = n - \operatorname{rank}(C).$$
⁽²⁹⁾

Definition 8. For a matrix Im(B) of dimension $n \times m$, the following equation is defined:

$$\operatorname{Im}(B) = \{ x = Bu \mid u \in \mathcal{U} \}.$$
(30)

Im (B) is a subspace of $\mathcal{X} = \mathbb{R}^n$. This subspace can be obtained with the following equation:

$$\dim \operatorname{Im}(B) = \operatorname{rank}(B). \tag{31}$$

Definition 9. For a matrix A of dimension $n \times n$, Equation (32) shows the inverse image of the subspace S in A:

$$A^{-1}\mathcal{S} = \{ x \in \mathcal{X} \mid Ax \in \mathcal{X} \}.$$
(32)

 $A^{-1}S$ is a subspace of $\mathcal{X} \subseteq \mathbb{R}^n$: For this subspace, dim $(A^{-1}S)$ is solved with the following equation:

$$\dim(A^{-1}\mathcal{S}) = \dim(\ker(A)) + \dim(\mathcal{S} \cap \operatorname{Im}(A)).$$
(33)

Definition 10 ([43]). A subspace $\mathcal{V} \subseteq \mathbb{R}^n$ is said to be A-invariant if $A\mathcal{V} \subset \mathcal{V}$; that is, if every state path x begins in \mathcal{V} and does not leave \mathcal{V} .

Definition 11 ([43]). A subspace $\mathcal{V} \subseteq \mathbb{R}^n$ is said to be $(A, \operatorname{Im}(B))$ -invariant if there exists $F: \mathcal{X} \to \mathcal{X}$ defined as Fx = u; that is, \mathcal{V} is a subspace $(A, \operatorname{Im}(B))$ -invariant if every trajectory of state x initiated in \mathcal{V} achieves to keep in \mathcal{V} bound by an appropriate control law u (see Figure 7), such that

$$(A+BF)\mathcal{V}\subset\mathcal{V}.\tag{34}$$

From Equation (34), *the gains of F are obtained.*



Figure 7. (a) Controlled invariants; (b) feedback control system.

Remark 1. If \mathcal{V} are $(A, \operatorname{Im}(B)$ -invariant, then also \mathcal{V} + is $(A, \operatorname{Im}(B))$ -invariant. For a subspace $\mathcal{P} \subset \mathcal{X}$, there exists a supreme subspace $(A, \operatorname{Im}(B))$ -invariant contained in \mathcal{P} . This supreme subspace can be calculated as the limit of the following non-increasing algorithm [44]:

$$\mathcal{P}^{0} = \mathcal{P},$$

$$\mathcal{P}^{i+1} = \mathcal{P} \cap A^{-1}(\mathcal{P}^{i} + \operatorname{Im}(B)).$$
(35)

Lemma 1. The subspace $\mathcal{V} \subset \mathcal{X}$ is $(A, \operatorname{Im}(B))$ -invariant if and only if

Proof. Suppose that \mathcal{V} is $(A, \operatorname{Im}(B))$ -invariant. Let $v_1 \in \mathcal{V}$; then,

$$(A + BF)v_1 = v_2 \in \mathcal{V},$$

$$Av_1 = v_2 - BFv_1 \in \mathcal{V} + \operatorname{Im}(B).$$
(36)

Then,

$$A\mathcal{V} \subset \mathcal{V} + \operatorname{Im}(B). \tag{37}$$

Now, suppose that $AV \subset V + \text{Im}(B)$. Set $\{e_1, \ldots, e_k\}$ as a basis of V, and then $Ae_i = v_i + b_i$ for $i = 1, \ldots, k$, where $v_i \in V$, $b_i = Bu_i$. Set $F : \mathcal{X} \to \mathcal{U}$ such that $Fe_i = u_i$.

$$Ae_i = v_i + b_1,$$

= $v_1 + Bu_i,$
= $v_1 + -BFe_i.$ (38)

Then,

$$Ae_i - BFe_i = v_i,$$

(A - BF)e_i = v_i \in V. (39)

Therefore, $(A + BF)\mathcal{V} \subset \mathcal{V}$. \Box

Definition 12. *If state* x *is measurable, then the disturbance rejection problem has a solution and there is a control law* u = Fx *such that*

$$\langle A + BF \mid \operatorname{Im}(E) \rangle \subset \operatorname{Ker}(C),$$
 (40)

where $\langle A + BF | \text{Im}(E) \rangle = \text{Im}(E) + (A + BF)\text{Im}(E) + (A + BF)^2\text{Im}(E) + ... + (A + BF)^{n-1}\text{Im}(E).$

Let us denote $\Omega = A + BF$.

Now, substituting u = Fx in the system, (A, B, C, D) is obtained:

$$\dot{x} = (A + BF)x + Ed, \quad y = Cx. \tag{41}$$

Theorem 1. The disturbance rejection problem has a solution if and only if

$$\operatorname{Im}(E) \subset \mathcal{V}^*, \tag{42}$$

where \mathcal{V}^* is the supreme of the family of subspaces $(A, \operatorname{Im}(B))$ – invariants contained in ker (C).

Proof. Suppose that the disturbance rejection problem has a solution. Set

$$\langle \Omega \mid \operatorname{Im}(E) = \mathcal{V} \rangle,$$
 (43)

where $\Omega = A + BF$. That is, Im $(E) + \Omega$ Im $(E) + \Omega^2$ Im $(E) + ... + \Omega^{n-1}$ Im $(E) = \mathcal{V}$. Therefore, the equation would be

$$\operatorname{Im}(E) \subset \mathcal{V}.$$
(44)

Since $(A + B)V \subset V$, then V belongs to the family of subspaces (A, Im B)-invariant contained in ker (C) [45,46]. Therefore,

 $\operatorname{Im}(E) \subset \mathcal{V} \subset \mathcal{V}^*. \tag{45}$

Now, suppose that $\text{Im}(E) \subset V^*$; then,

$$\langle \Omega \mid \mathrm{Im}(E) \rangle \subset \langle \Omega \mid \mathcal{V}^* \rangle.$$
 (46)

As $\langle \Omega \mid \mathcal{V}^* \rangle = \mathcal{V}^* + \Omega \mathcal{V}^* + ... + \Omega^{n-1} \mathcal{V}^* \subset \mathcal{V}^* \subset \text{Ker}(C)$, then $\langle \Omega \mid \text{Im}(E) \subset \text{Ker}(C) \rangle$.

 \mathcal{V}^* can be calculated as the limit of the following non-increasing algorithm.

$$\mathcal{V}^{0} = \operatorname{Ker}(C),$$

$$\mathcal{V}^{j+1} = \operatorname{Ker}(C) \cap A^{-1}(\operatorname{Im}(B) + \mathcal{V}^{j}).$$
(47)

When j = 0, 1, 2, 3, ... Then, $\mathcal{V}^{j+1} \subset \mathcal{V}^j$ and for some $l \leq \dim \ker (C)$ is true for all $l \geq k$:

$$\mathcal{V}^{j} = \mathcal{V}^{\star}. \tag{48}$$

where \mathcal{V}^* is the supreme subspace (*A*, Im(*B*))-invariant contained in ker (*C*). \Box

Definition 13. *If the state x and the disturbance d are measurable, then the disturbance rejection problem has a solution if there is a control law u = Fx + Gd (see Figure 8), such that*

$$\operatorname{Im}(E) \subset \mathcal{V}^* + \operatorname{Im}(B). \tag{49}$$

Now, substituting u = Fx + Gd into the system, (A, B, C, E) is obtained:

$$\dot{x} = (A + BF)x + (BG + E)d, \ y = Cx.$$
 (50)



Figure 8. Geometric control scheme applied to an LED driver using a buck converter.

Proposition 1. *Im* $(E) \subset \mathcal{V}^*$ *if and only if Im* $(E) \subset \mathcal{V}^* + Im$ (B)*:*

Proof. Suppose Im $(E) \subset \mathcal{V}^*$; then,

$$\begin{aligned} &(E)v_1 = v_2 \in \mathcal{V}^*, \\ &Ev_1 = v_2 \in \mathcal{V}^*. \end{aligned}$$
 (51)

Therefore,

$$\operatorname{Im}(E) \subset \mathcal{V}^* + \operatorname{Im}(B).$$
(52)

Now, set $\{e_1, ..., e_k\}$ as the basis of \mathcal{V}^* , and then $Ee_i = v_i + b_i$, i = 1, ..., k, where $v_i \in \mathcal{V}^*$, $b_i = Bu_i$. Define $G : \mathcal{D} \to \mathcal{U}$ such that $Ge_i = u_i$; then,

$$Ee_i = v_i + b_i,$$

= $v_i + Bu_i,$
= $v_i + BGe_i.$ (53)

So,

$$\begin{aligned} Ee_i &= v_i, \\ (E)e_i &= v_i \in \mathcal{V}^*. \end{aligned}$$
(54)

Then,

$$\operatorname{Im}(E - BG) \subset \mathcal{V}^*.$$
(55)

From Equation (55), the gain of *G* is obtained. \Box

After defining the theory and design of the robust controllers (structure-at-infinity control and geometric control), we continue to design a precompensated system since this robust controller scheme only works correctly to stabilize and carry the output to zero. Therefore, it is necessary to have trajectory tracking with non-zero input values and to be able to control the output y(t). This is described in the next subsection.

3.3. Precompensated System

The robust controllers only have regulation of the capacitor voltage and inductor current outputs, and it is necessary to implement a precompensated part to track the desired trajectories and achieve more robust control with good performance in different scenarios.

Therefore, since a non-zero reference must be followed, there is a compensation to improve the steady-state error.

From the above, a precompensation gain *N* needs to be implemented, so the closed-loop transfer function (*cl*) is as follows:

$$G_{cl}(s) = \frac{Y(s)}{U(s)} = NC[sI - (A - BK)^{-1}]B$$

= $N \frac{b_m s^m + b_{m-1} s^{m-1} + \dots + b_1 s + b_0}{s^n + \alpha_{n-1} s^{n-1} + \dots + \alpha_1 s + \alpha_0}, m < n.$ (56)

Considering that the state feedback acts on the poles and not on the zeros, when r = 0, the system's evolution depends only on the poles, and, when $r \neq 0$, the zeros also participate in the system's response.

Applying the limit to Equation (56), the following equation is obtained:

$$y_{ss} = \lim_{t \to \infty} y(t) = G_{cl}(0) = N \frac{b_0}{\alpha_0} \Rightarrow N = \frac{\alpha_0}{b_0},$$

$$y_{ss} = \lim_{t \to \infty} y(t) = -NC[A - BF]^{-1}B = 1.$$
(57)

The precompensation gain is

=

$$N = -\frac{1}{C[A - BF]^{-1}B}.$$
(58)

From the mathematical theory of structure-at-infinity control and geometric control, the precompensation system for tracking trajectories is implemented, resulting in the following control functions:

Structure at infinity control:

$$Fault Compensation \Rightarrow u = Fx + Jk + Nr,$$

Disturbance rejection
$$\Rightarrow u = Fx + Jd + Nr.$$
 (59)

Geometric control:

Fault Compensation
$$\Rightarrow u = Fx + Gk + Nr$$
,
Disturbance rejection $\Rightarrow u = Fx + Gd + Nr$. (60)

These equations are applied to the LED driver using a buck converter to stabilize the system, attenuate disturbances, fault compensation, and follow desired trajectories.

Subsequently, in the next section, the implementation of the controllers on the systems with disturbance and fault will be carried out, to attenuate and compensate for unwanted inputs and faults in the actuator.

4. Disturbance Attenuation and Fault Compensation through Structure-at-Infinity Control and Geometric Control

This section presents the analytical development of the controllers applied to the LED driver using a buck converter.

The applied controllers aim to stabilize and follow the trajectories, compensating for the effect of faults in the actuator and attenuating the disturbances at the electrical systems' input. This is completed to ensure good performance and robustness of the LED system with a buck converter.

First, the development and implementation of the structure-at-infinity control will be carried out. This is described in the following subsection.

4.1. *Compensation of the Effect of a Fault in the Actuator via Structure at the Infinity Control* Returning to the system in state-space of Equation (8), the system is as follows:

$$\begin{bmatrix} \dot{x}_1 \\ \dot{x}_2 \end{bmatrix} = \begin{bmatrix} 0 & -193.42 \\ 2083000 & -92428.27 \end{bmatrix} \begin{bmatrix} x_1 \\ x_2 \end{bmatrix} + \begin{bmatrix} 15473.88 \\ 0 \end{bmatrix} u + \begin{bmatrix} 15473.88 \\ 0 \end{bmatrix} k + \begin{bmatrix} 0 \\ 92428.27 \end{bmatrix}, y = \begin{bmatrix} 0 & 1 \end{bmatrix} \begin{bmatrix} x_1 \\ x_2 \end{bmatrix},$$
(61)

where *R* is a constant input to the plant and the fault is *Bk*.

Following the steps defined by the theory shown in Section 3.1, we first develop the transfer functions of the system without fault (A, B, C) of Equation (19) and with fault (A, B, C, E) of Equation (20), and it was determined that they are equal; therefore, through the structure in infinity control, the effect of the fault can be compensated. This is determined by Equation (27). Rewrite H(s) and $H^d(s)$ as follows:

$$H(s) = H^{d}(s) = \frac{s(1803.2)}{s^{2}(5.59 \times 10^{-8} + 0.00517s^{-1} + 22.54s^{-2})} = \frac{1}{s}g(s).$$
(62)

Solving for g(s) from Equation (62),

$$g(s) = g^{d}(s) = \frac{1803.2}{5.59 \times 10^{-8} + 0.00517s^{-1} + 22.54s^{-2}}.$$
(63)

Calculating lim,

$$\lim_{s \to \infty} g(s) = \lim_{s \to \infty} g^d(s) = \lim_{s \to \infty} \frac{1803.2}{5.59 \times 10^{-8} + 0.00517s^{-1} + 22.54s^{-2}} = 32.26 \times 10^9 \neq 0.$$
(64)

Since $\lim_{s\to\infty} g(s) \neq 0$ and $\lim_{s\to\infty} g^d(s) \neq 0$, then H(s) and $H^d(s)$ has a zero at the infinity of order 1; therefore, a control function can be found that compensates for the effect of the fault on the actuator of the system.

The function u = Fx + Jk that compensates for the effect of the fault on the system is obtained by substituting the values in $J(s) = C(sI - A - BF)^{-1}(BJ + E)$.

Here, $F = \begin{bmatrix} f_1 & f_2 \end{bmatrix}$; these are calculated using pole assignment using Equation (16). The characteristic polynomial P_c of the system is

$$P_c = s^2 + (92428.27 + 15474f_1)s + (1.4302 \times 10^9 f_1 + 3.2232 \times 10^{10} f_2 + 4.0289 \times 10^8).$$
(65)

Based on Equation (A6) that describes the underdamped second-order system, the desired polynomial is defined:

$$P_d = s^2 + 2\zeta\omega_n s + \omega_n^2 = s^2 + 9.2427 \times 10^4 s + 3.1296 \times 10^9.$$
(66)

Equating the coefficients of P_c and P_d , we obtain

$$f_1 = -7.361 \times 10^{-5},$$

$$f_2 = 0.0845.$$
(67)

Then,

$$F = \begin{bmatrix} -7.361 \times 10^{-5} & 0.0845 \end{bmatrix}.$$
(68)

Once the gain matrix *F* has been determined, it is necessary to find the parameters of *J* and be able to check the control function u = Fx + Jk that compensates for the effects of fault on the actuator; for this, it is necessary that $J(s) = C(sI - A - BF)^{-1}(BJ + E) = 0$; therefore,

$$J(s) = \frac{1803.2(J+1)}{5.59 \times 10^{-8}s^2 + 5.17 \times 10^{-3}s - 129.82} = 0.$$
 (69)

Since the function J(s) = 0, we can now obtain the value of J from Equation (69):

$$1803.2(J+1) = 0,$$

$$J = -1.$$
(70)

From Equation (58), *N* was calculated and is obtained:

$$N = 0.0971.$$
 (71)

So, the function u = Fx + Jk + Nr that compensates for the effect of the fault on the system is

$$u = \begin{bmatrix} -7.361 \times 10^{-5} & 0.0845 \end{bmatrix} x - k + 0.0971r.$$
(72)

After having found the control law (structure-at-infinity control) to compensate for the fault, we proceed to find the geometric control function to compensate for the fault; this is described in the next subsection.

4.2. Compensation of the Effect of a Fault in the Actuator via Geometric Control

For this case, the same system with fault obtained from Equation (61) was presented to compare the robust controllers.

For the analytical development and implementation, the steps defined by the geometric control theory were followed, which is shown in Section 3.2.

First, the invariant subspace \mathcal{V}^* was determined, starting with $\mathcal{V}^0 = \ker(C)$. The augmented matrix *C* is defined as follows:

The augmented matrix *C* is defined as follows:

$$C = \begin{bmatrix} 0 & 1 \mid 0 \end{bmatrix}. \tag{73}$$

The reduced stepped form is calculated to obtain the following equation.

$$x_{2} = 0$$

$$x = \begin{bmatrix} x_{1} \\ x_{2} \end{bmatrix} = x_{1} \begin{bmatrix} 1 \\ 0 \end{bmatrix} + 0.$$
(74)

Therefore,

$$\mathcal{V}^{0} = gen\left\{ \begin{bmatrix} 1\\ 0 \end{bmatrix} \right\} = \ker(C).$$
(75)

Now, it is defined as follows:

$$\operatorname{Im}(B) = gen\left\{ \begin{bmatrix} 1\\ 0 \end{bmatrix} \right\}.$$
(76)

Then,

$$\operatorname{Im}(B) + \mathcal{V}^0 = gen\left\{ \begin{bmatrix} 1\\ 0 \end{bmatrix} \right\}.$$
(77)

It is observed that

$$\mathcal{V}^{1} = \ker(C) \cap A^{-1}(\operatorname{Im}(B) + \mathcal{V}^{0}),$$

$$\mathcal{V}^{1} = gen\left\{ \begin{bmatrix} 1\\ 0 \end{bmatrix} \right\}.$$
(78)

So,

$$\mathcal{V}^* = gen\{\mathcal{V}_1\} = gen\left\{\begin{bmatrix}1\\0\end{bmatrix}\right\}.$$
(79)

To verify that Im(E) is contained in $\mathcal{V}^* + Im(B)$, it is necessary to know Im(B):

$$\operatorname{Im}(B) = \begin{cases} 1\\0 \end{cases} = (1) \begin{bmatrix} 1\\0 \end{bmatrix} + (0) \begin{bmatrix} 0\\1 \end{bmatrix}.$$
(80)

Since the previous criterion $(\text{Im}(E) \subset (V^* + \text{Im}(B)))$ was met, it is now necessary to calculate u = Fx + Gd.

Then, there exist $F = \begin{bmatrix} f_1 & f_2 \end{bmatrix}$ and G such that u = Fx + Gd compensates for the effect of the fault; therefore, it is necessary to find the values of F such that

$$(A+BF)\mathcal{V}^* \subset (\mathcal{V}^* + \operatorname{Im}(B)).$$
(81)

Then,

$$(A+BF)\mathcal{V}^* = \begin{bmatrix} 15473.88f_1 & 193.42 + 15473.88f_2\\ 2.083 \times 10^6 & 92428.27 \end{bmatrix} \left\{ \begin{bmatrix} 1\\0 \end{bmatrix} \right\}.$$
 (82)

Likewise,

$$f_1 = \begin{bmatrix} 15473.88f_1\\ 2.083 \times 10^6 \end{bmatrix} = \begin{bmatrix} 1\\ 0 \end{bmatrix}, f_2 = 0.$$
(83)

Then, the value of F is obtained by solving the following system of equations; therefore,

$$F = \begin{bmatrix} 6.4625 \times 10^{-5} & 0 \end{bmatrix}. \tag{84}$$

Substituting *F* into $C(sI - A - BF)^1(BG + E) = 0$ is obtained *G*, so

$$G(s) = \frac{1803.2(G+1)}{s^2 + 5.17 \times 10^{-3}s + 22.53} = 0,$$

$$1803.2(G+1) = 0,$$

$$G = -1.$$
(85)

The value of *N* obtained is equal to that shown in Equation (71). The function u = Fx + Gk + Nr via geometric control that compensates for the effect of the fault on the actuator is

$$u = \begin{bmatrix} 6.4625 \times 10^{-5} & 0 \end{bmatrix} x - k + 0.0125r.$$
(86)

After having found the geometric control function to compensate for the fault, we now proceed with the disturbance rejection case using the two robust control laws. In the next section, the development and implementation of the structure-at-infinity control will be carried out to mitigate disturbances in the electrical system.

4.3. Rejection and Attenuation of the Disturbance on the Power Supply via Structure at the Infinity Control

In this case, the control theory described in Section 3.1 must be followed.

For the case of a power supply disturbance, Equation (10) is applied, resulting in the following equation:

$$\begin{bmatrix} \dot{x}_1 \\ \dot{x}_2 \end{bmatrix} = \begin{bmatrix} 0 & -193.42 \\ 2083000 & -92428.27 \end{bmatrix} \begin{bmatrix} x_1 \\ x_2 \end{bmatrix} + \begin{bmatrix} 15473.88 \\ 0 \end{bmatrix} u + \begin{bmatrix} 95.745 \\ 0 \end{bmatrix} d + \begin{bmatrix} 0 \\ 92428.27 \end{bmatrix}, y = \begin{bmatrix} 0 & 1 \end{bmatrix} \begin{bmatrix} x_1 \\ x_2 \end{bmatrix}.$$
(87)

Following the steps (Section 3.1) first, the transfer functions $H(s) = C(sI - A)^{-1}B$ and $H^d(s) = C(sI - A)^{-1}E$ for this case are not similar and are defined as follows. Therefore, Equations (19) and (20) are defined as follows:

$$H(s) = \frac{s(1803.2)}{s^2(5.59 \times 10^{-8} + 0.00517s^{-1} + 22.54s^{-2})} = \frac{1}{s}g(s),$$
(88)

$$H^{d}(s) = \frac{s(11.1573)}{s^{2}(5.59 \times 10^{-8} + 0.00517s^{-1} + 22.54s^{-2})} = \frac{1}{s}g_{d}(s),$$
(89)

where, by solving for g(s) and $g^{d}(s)$, the following equations are obtained:

$$g(s) = \frac{1803.2}{5.59 \times 10^{-8} + 0.00517s^{-1} + 22.54s^{-2}},$$
(90)

$$g^{d}(s) = \frac{11.1573}{5.59 \times 10^{-8} + 0.00517s^{-1} + 22.54s^{-2}}.$$
(91)

Calculating lim,

$$\lim_{s \to \infty} g(s) = \lim_{s \to \infty} \frac{1803.2}{5.59 \times 10^{-8} + 0.00517s^{-1} + 22.54s^{-2}} = 32.26 \times 10^9 \neq 0, \tag{92}$$

$$\lim_{s \to \infty} g^d(s) = \lim_{s \to \infty} \frac{11.1573}{5.59 \times 10^{-8} + 0.00517s^{-1} + 22.54s^{-2}} = 19.96 \times 10^7 \neq 0.$$
(93)

Since $\lim_{s\to\infty} g(s) \neq 0$ and $\lim_{s\to\infty} g^d(s) \neq 0$, then H(s) and $H^d(s)$ has a zero at the infinity of order 2, so it is possible to find a control function that attenuates or rejects the disturbance in the system.

The function u = Fx + Jd that rejects the disturbance in the system is obtained by substituting the values in $J(s) = C(sI - A - BF)^{-1}(BJ + E)$.

The gains *F* are then calculated from the Equation (34); then,

$$F = \begin{bmatrix} -7.361 \times 10^{-5} & 0.0845 \end{bmatrix}.$$
(94)

Once the gain matrix *F* has been determined, it must be verified that the function u = Fx + Jd that attenuates the disturbances, as long as $J(s) = C(sI - A - BF)^{-1}(BJ + E) = 0$; for this, the values of *J* must be found. Therefore,

$$J(s) = \frac{22.54(80J + 0.495)}{5.59 \times 10^{-8}s^2 + 5.22 \times 10^{-3}s - 129.82} = 0.$$
 (95)

Since the function J(s) = 0, the value of *J* can be obtained from Equation (95):

$$22.54(80J + 0.495) = 0,$$

$$J = -0.00619.$$
(96)

The value of *N* obtained is equal to that shown in Equation (71). Therefore, the control function u = Fx + Jd + Nr that rejects the disturbance in the system is

$$u = \begin{bmatrix} -7.361 \times 10^{-5} & 0.0845 \end{bmatrix} x - 0.00619d + 0.0971r.$$
(97)

After having found the control law (structure-at-infinity control), the development and implementation of geometric control continue with the objective of finding the control law that rejects and attenuates disturbances. This is described in the following subsection.

4.4. Rejection and Attenuation of the Disturbance on the Power Supply via Geometric Control

For this case, the same perturbed state-space representation is used, which is shown in Equation (87):

Following the steps of the control theory shown in Section 3.2, first, the same values are taken from the Equation (84), which represent the gains of f, where $F = \begin{bmatrix} f_1 & f_2 \end{bmatrix}$; therefore,

$$F = \begin{bmatrix} 6.4625 \times 10^{-5} & 0 \end{bmatrix}. \tag{98}$$

The next step is to substitute *F* into $C(sI - A - BF)^1(BG + E) = 0$ to obtain *G*; then,

$$G(s) = \frac{22.54(80G + 0.495)}{s^2 + 5.17 \times 10^{-3}s + 22.53} = 0,$$

$$80G + 0.495 = 0,$$

$$G = -0.00619.$$
(99)

The value of *N* obtained is equal to that shown in Equation (71). The function u = Fx + Gd + Nr via geometric control that rejects the disturbance in the system is

$$u = \begin{bmatrix} 6.4625 \times 10^{-5} & 0 \end{bmatrix} x - 0.00619d + 0.0125r.$$
(100)

After obtaining the analytical results of the two robust controllers applied to reject disturbances and compensate for faults in the LED driver, we continue to the results and discussion.

5. Results and Discussion

This section discusses the results obtained with the different control scenarios using structure-at-infinity control and via geometric control.

Table 2 summarizes the control parameters obtained by analyzing each scenario mentioned in Section 4.

Control parameters with actuator fault (MOSFET)							
Geometric control			Via structure at the infinity control				
F	G	Ν	F	J	Ν		
$[6.462 \times 10^{-5} 0]$	-1	0.0125	$[-7.36 imes 10^{-5} \ 0.0845]$	-1	0.0971		
Control parameters with disturbance in power supply							
Geometric control		Via structure at	Via structure at the infinity control				
F	G	Ν	F	J	Ν		
$[6.462 \times 10^{-5} 0]$	-0.00619	0.0125	$[-7.36 imes 10^{-5} \ 0.0845]$	-0.00619	0.0971		

Table 2. Parameters of control via structure at the infinity control and geometric control.

The first scenario shown uses a structure-at-infinity control to stabilize, track the desired trajectory, and compensate for the effect of the fault presented in the system. This is shown in the following subsection.

Compensation of actuator fault using structure-at-infinity control.

For this first scenario, a sine wave-type fault is applied in the actuator (MOSFET), as shown in Figure 3a. In this figure, it can be observed how the structure-at-infinity control performs when the compensation term, *Jk*, is considered or not, as well as the output signals of the capacitor voltage and inductor current of the electrical system (LED driver using a buck converter).

Figure 9a shows the control signals generated by the PWM to keep the system stable and compensate for the effect of variable fault. It can be observed that the effort generated by the via structure at the infinity control varies between 0.29 and 0.63, resulting in a more aggressive signal with high oscillations to compensate for the effect of the sine wave-type fault. For the controller without fault compensation, a soft signal is observed with values ranging from 0.32 to 0.6 to follow the desired path.

Likewise, Figure 9b,c show the output responses (capacitor voltage and inductor current) obtained by the via structure at the infinity control, control without compensation for faults, and without control. When using the via structure at the infinity control, it can be seen in Figure 9b that, in the first seconds, a voltage overimpulse is generated that goes from 39 v to 42 v, subsequently achieving stability at the desired reference of 40 v (this voltage value has been set since the plant operates up to 80 v, but the system has a duty cycle of 50% and operates at 40 v). Due to the presence of the sine wave-type fault and the desired voltage change, the structure-at-infinity control again loses its trajectory at those moments of time (4 \times 10⁻³ s); however, it quickly recovers the desired voltage again. Control actions also impact the inductor current, which tends to have an overimpulse in the first seconds ranging from 0.4 A to 0.3 A; however, with these parameters, the LED driver can function properly (see Figure 9c). In the case of a controller without fault compensation, it is observed that it has oscillations ranging from 30 to 46 volts, causing it to lose the desired voltage of 40 volts at instants of time $(1 \times 10^{-3} \text{ s to } 4 \times 10^{-3} \text{ s})$, and the MOSFET is affected by the oscillations generated by the fault. Consequently, the inductor current is also affected, generating values ranging from 0.5 A to 0.2 A, causing a critical state for the correct operation of the LED driver using a buck converter.

The changes in the established trajectories range from 0 v to 40 v and 40 v to 35 v; the generated change was 4×10^{-3} s. It is observed that there is an error difference of 5 v between the output capacitor voltage obtained (using the controller without fault compensation) and the defined reference. It is important to mention that the output voltage obtained varies from 30 v to 46 v, which is why it is a significant loss for the LED driver using a buck converter; therefore, the desired voltage is not reached after 7×10^{-3} s in the presence of a sine wave-type fault.

On the other hand, we can observe that, when the electrical circuit does not have control, output voltages ranging from 0 v to 46 v and a current that oscillates from 0 A to

0.65 A are obtained, resulting in the overload and loss of functionality of the LED driver using a buck converter (see Figure 4).

Subsequently, the simulation results of the geometric control that compensates for the actuator fault are shown in the following subsection.

• Compensation of actuator fault through geometric control.

On the other hand, when geometric control is used, the results obtained are different, with a great contribution to the LED driver using a buck converter.

The fault generated in the system (see Figure 3a) is the same as that applied with the via structure at the infinity control; however, the control signal presented with the geometric control is more aggressive, with PWM values that range between 0.25 and 0.68 (see Figure 10a); this causes the controller effort to be greater compared to the via structure at the infinity control.

Figure 10b,c show the voltage of the capacitor and the current of the inductor. In Figure 10b, it is observed that the voltage obtained follows the desired path and varies from 40 v to 35 v with a margin of error of 1 v. Similarly, the current tends to follow the path, generating values ranging from 0.3 A to 0.18 A. These parameters are within the appropriate values for the LED driver that uses a buck converter to function properly. It is important to mention that the geometric control does not generate overimpulse in the first seconds, as shown in the results of the structure at the infinite control.

The results shown using the geometric control make a great contribution since it is observed that the fault generated in the actuator is compensated; however, the control signal has a similar tendency to the sine wave-type fault signal, particularly because it does not affect the LED driver using a buck converter since both the capacitor voltage and the inductor current follow the desired path and do not present alterations, resulting in good performance and robustness against actuator faults.

However, the results obtained with control without fault compensation and without control present the same dynamics as in the previous scenario, providing results of high voltages and currents outside the operating range that can affect the LED driver using a buck converter.

After observing the simulation results, we continue with the study and analysis of the simulation results for the case of disturbance rejection using robust controllers, with the objective of comparing and defining the best performance and robustness in these scenarios that can affect the operation of the LED driver using a buck converter.

 Rejection and attenuation of disturbance in the power supply through structure-atinfinity control.

For this scenario, the results obtained from a power supply disturbance are presented (see Figure 3a); this signal manifests itself as a malfunction defined as an unwanted input.

For this scenario, the control signal generated via the structure-at-infinity control tends to have PWM variations ranging from 0.85 to 0.085 (see Figure 11a); this causes the capacitor voltage to go from 0 to 43 v and from 40 v to 33 v with an error of 2 v between the desired and obtained voltages (see Figure 11b). Consequently, it is observed in Figure 3c how the inductor current is minimally affected by the disturbance, resulting in current values ranging from 0.30 A to 0.13 A. As a result of these signals, the LED driver using a buck converter will work correctly, and, consequently, it would not have a higher risk of generating a short circuit or an overloaded circuit (Figure 11c).

For the case that has the greatest impact on the system, the results obtained with control without disturbance attenuation present a control signal that cannot attenuate the disturbances generated by the ramp- and step-type signals, causing the PWM values to go from 0.39 to 0.34. Likewise, the results obtained for voltage (0.30 v) and current (-0.15 A) are lower than the desired trajectory, resulting in the LED driver using a buck converter having a malfunction and consequently a greater risk of generating a short circuit or an overloaded circuit.



Figure 9. (a) Control signals to compensate for fault in the actuator (MOSFET); (b) capacitor voltage outputs using the via structure at infinity control; (c) inductor current outputs using the via structure at infinity control.



Figure 10. (**a**) Control signals to compensate for fault in the actuator (MOSFET); (**b**) capacitor voltage outputs using geometric control; (**c**) inductor current outputs using geometric control.



Figure 11. (a) Control signals to attenuate and reject disturbances in the power supply; (b) capacitor voltage outputs using the via structure at infinity control; (c) inductor current outputs using the via structure at infinity control.

On the other hand, we can observe the behavior of the capacitor voltage and the inductor current without control (see Figure 4). The output signals obtained are not similar to the control without disturbance attenuation, but the result is the same since, with those values of voltage (0 v to 44 v) and current (0 A to 0.4 A), the LED driver using a buck converter will not have the correct power and an overvoltage will be generated.

For the fourth scenario, the via geometric control is presented with the same disturbance tests on the LED driver using a buck converter.

• Rejection and attenuation of disturbance in the power supply through geometric control.

In this scenario, the ramp- and step-type disturbance shown in Figure 3b is presented. As mentioned above, these data affect the voltage input, generating an overvoltage. Figure 12a shows a stable signal (PWM) from the geometric control. The effort generated by the controller is smooth, with oscillations ranging from 0.5 to 0.4 v.



Figure 12. (**a**) Control signals to attenuate and reject disturbances in the power supply; (**b**) capacitor voltage outputs using geometric control; (**c**) inductor current outputs using geometric control.

Figure 12a shows the effort generated with geometric control and with control without disturbance attenuation. The generated control signals are similar; however, the geometric control signal presents greater effort compared to the controller without disturbance attenuation. For the controller without disturbance attenuation, the signals are softer, with PWM values ranging from 1 to 0.4.

From the PWM signals generated by the controllers, the capacitor voltage and inductor current signals were obtained as output. Figure 12b shows that, by using geometric control, a stable voltage is obtained that is not affected by the disturbance that was generated in the power supply. It is possible to reach the established references with a quick response or with adequate response times that allow the desired voltage to be reached. In the case of the voltage output response using the control without disturbance attenuation, it is observed that it presents increases ranging from 43 v to 39.6 v, in which these voltage peaks can generate a malfunction in the LED driver using a buck converter.

Consequently, the same dynamic behavior generated by the disturbances and the signals generated by the controllers occurs in the inductor current (see Figure 12c), generating currents that oscillate (0.37 A to 0.25 A) due to the effect of the disturbance.

In the fault scenario, it can be observed that the structure-at-infinity control presents an overimpulse in the first milliseconds with an error of 0.025%; subsequently, it stabilizes the system for 4×10^{-3} s and achieves following the trajectory after 1×10^{-4} s. The opposite is true for geometric control, which does not present overimpulse but does present a time greater than 4×10^{-4} s to follow the desired trajectory again. However, both controllers work properly as the values are appropriate for the operation of the LED driver.

For the disturbance scenario, it was observed that the structure-at-infinity control presents an overimpulse in the first milliseconds; however, it quickly achieves following the trajectory for 4×10^{-3} s despite the two disturbances presented in that period. Subsequently, the change in the trajectory is presented, but it has a greater error compared to the geometric control, which presents a better performance without overimpulse and an approximation with an error of 0.05%.

Both controllers provide rapid responses to stabilize, reject disturbances, and compensate for faults and path changes without affecting the LED driver.

6. Conclusions

Using the model of the LED driver using a buck converter, it was possible to implement two controllers (structure-at-infinity control and geometric control) with great performance against the trajectory changes and robustness against the MOSFET faults and disturbances in the power supply.

In the first scenario, it is concluded that both controllers have great performance; however, the via structure at the infinity control presents a greater error of 1 v between the reference and the voltage of the capacitor, and this difference is generated when a fault occurs, generating a malfunction in the electrical circuit. In the case of the geometric control, the performance is outstanding since it follows the desired voltage (40 v and 35 v) and compensates for the effect of the fault generated on the MSOFET, resulting in greater robustness in better response times, in compliance with the operation duty cycle of the LED driver using a buck converter. Likewise, it is concluded that both the control signal and the voltage and current output data are better when using geometric control than regarding the via structure at the infinity control since the oscillations generated by the fault are compensated, obtaining more stable responses without alterations, with an error of 0.05 v.

For the second scenario, it is concluded that the geometric control and structureat-infinity control are robust to disturbances, achieving attenuation or rejecting them. However, the established desired voltages can be approached and quickly followed by the geometric control, and, when the via structure at the infinity control is used, the system remains stable but tends to overshoot; however, it achieves stabilization and reaches the desired voltages. Finally, it can be concluded that both controllers have better robust characteristics for the different scenarios as they quickly follow the desired trajectory, compensate for the effect of the fault, and attenuate the disturbances at the system input. With these controllers, the overvoltage and degradation in the capacitor and inductor can be avoided, ensuring proper operation.

In future work, experimental tests are required to implement geometric control with robustness tests on faults and disturbances. Likewise, it is necessary to conduct additional comparisons with other robust controllers, such as Fault-Tolerant MPC or Fault-Tolerant Control.

Author Contributions: Conceptualization, M.A.J. and J.E.V.-R.; Methodology, G.O.-T. and J.A.B.-M.; Validation, H.M.B.-A. and M.R.-M.; Formal analysis, J.G.-R.; Investigation, J.Y.R.-M., M.A.L.-O., C.A.T.-C. and M.C.-R.; Writing—review & editing, F.D.J.S.-V. All authors have read and agreed to the published version of the manuscript.

Funding: This research and the APC was funded by CONAHCYT grant number CF-2023-G-1344.

Data Availability Statement: Data are contained within the article.

Acknowledgments: We Authors would like to thank CONAHCYT for their support through the project CF-2023-G-1344 Desarrollo de nuevos enfoques para analisis, modelado y control de convertidores electronics. Also, want to thank the Centro Universitario de los Valles of the University of Guadalajara for the access and use of the Laboratorio de Captura Óptica de Movimiento, and to the Consejo Nacional de Humanidades, Ciencias y Tecnologías (CONAHCYT) for the scholarship awarded to student Jair Gómez Radilla to carry out studies in the Maestría en Ciencias Físico Matemáticas con orientación en matemáticas.

Conflicts of Interest: The authors declare that they have no conflicts of interest.

Appendix A

Figure 1 presents the schematic of a buck converter that has a power load and an attached LED panel. As is well-known, a buck converter is a DC–DC converter with an output voltage lower than the one placed at its input and an output voltage value depending on the power switch (MOSFET) operation within the circuit. To model this power electronic dynamical system, a simplified LED model has been considered from [15,47], giving rise to the schematic presented in Figure A1.



Figure A1. LED model scheme.

As can be seen, the simplified model of an LED panel is replaced with a resistance R_L in series with a voltage source V_{ls} , allowing one to capture its behavior in terms of constant voltage drop across the LED and a resistor that limits current. The buck converter is composed of a MOSFET transistor (M), a diode (D), an inductor (L), and a capacitor (C). The MOSFET transistor is used as a controllable switch to regulate current flow, while the diode allows current to flow in a specific direction. As expected, the inductor and capacitor tasks are related to energy storage and transfer within the circuit. For this, let us point out that it is required, for efficiency matters, to keep the current through the LED regulated, giving rise to the control objective. In other words, the design of the closed-loop control system involves the current of the LED as a feedback signal to adjust and maintain it at

a desired value. This is achieved by modifying the useful duty cycle of the PWM signal that controls the MOSFET transistor. As the current of the LED deviates from the desired value, the controller adjusts the duty cycle to correct this deviation and keep the current constant. In addressing the mathematical model considered in this paper, we point out that, since the circuit operation is composed of two different behaviors depending on the MOSFET transistor state (active or inactive), an averaged model is required to be generated. To achieve this, let us focus on Figure A2, the schematic of the circuit operating in the case of active MOSFET.



Figure A2. Scheme of the LED model on.

In the case of MOSFET activation, note that diode D is reverse-polarized for the input voltage source V_{in} , and, consequently, no current flows through it. As a result, by applying the Kirchhoff law, the following is obtained:

$$i_L = i_C + i_{RL},$$

$$V_{in} = V_L + V_C.$$
(A1)

where i_L , i_C , and i_{RL} correspond to the flow current associated with the inductor L, capacitor C, and resistor R_L , respectively. In Equation (A1), V_L and V_C represent the voltages of the inductor and the capacitor, respectively. If the MOSFET is inactive or out of operation (see Figure A3), the stored current within the inductor L provides energy to the load.



Figure A3. Model scheme LED off.

As a result, the following can be stated:

$$V_C = V_{RL} + V_{ls},$$

$$i_L = i_C + i_{RL}.$$
(A2)

As such, the mathematical model must be derived taking into account the active elements within the circuit, which give rise to i_L and v_C as dynamical variables. As a result, notice that, since $i_C = C_c \frac{dV_C}{dt}$ and $V_L = L \frac{di_L}{dt}$ from Equation (A1):

$$i_{C} = i_{L} - i_{RL},$$

$$C_{c} \frac{dV_{C}}{dt} = i_{L} - i_{RL},$$

$$\frac{dV_{C}}{dt} = \frac{i_{L}}{C_{c}} - \frac{V_{RL}}{R_{L}C_{c}}.$$
(A3)

With *L* and *C_c* representing inductance and capacitance, respectively, recalling that $V_{RL} = V_C - V_{ls}$ from Equation (A2), Equation (A3) results in

$$\frac{dV_C}{dt} = \frac{i_L}{C_c} - \frac{V_C}{R_L C_c} + \frac{V_{ls}}{R_L C_c}.$$
(A4)

concluding with the differential equation that approximates the voltage of the capacitor. As for the value involved in i_L , notice that, from Equation (A1),

$$V_L = V_{in} - V_C,$$

$$L\frac{di_L}{dt} = V_{in} - V_C,$$

$$\frac{di_L}{dt} = \frac{V_{in}}{L} - \frac{V_C}{L}.$$
(A5)

Appendix B

The desired location of the closed-loop poles is chosen based on an underdamped secondorder system, with a maximum overshoot $M_p = 1\%$ and a settling time $t_s = 6.4916 \times 10^{-5}$ s ζ and ω_n calculated using the following equations:

$$\zeta = \frac{1}{\sqrt{(\frac{\pi}{\ln M_p})^2 + 1}}$$

$$\omega_n = \frac{3}{\zeta t_s}$$
(A6)

$$\zeta = \frac{\zeta}{\sqrt{(\frac{\pi}{\ln 0.01})^2 + 1}} = 0.8261$$
$$\omega_n = \frac{3}{(0.8261)(6.4916 \times 10^{-5})} = 5.5943 \times 10^4$$
$$P_d = s^2 + 2\zeta\omega_n s + \omega_n^2 = s^2 + 9.2427 \times 10^4 \text{ s} + 3.1296 \times 10^9$$

References

- 1. Safiullah, S.; Rahman, A.; Ahmad Lone, S. Optimal control of electrical vehicle incorporated hybrid power system with second order fractional-active disturbance rejection controller. *Optim. Control Appl. Methods* **2023**, *44*, 905–934. [CrossRef]
- Ren, H.R.; Ma, H.; Li, H.Y.; Lu, R.Q. A disturbance observer based intelligent control for nonstrict-feedback nonlinear systems. Sci. China Technol. Sci. 2023, 66, 456–467. [CrossRef]
- Rubio, J.; Figueroa, M.; Pérez Cruz, J.; Yoe Rumbo, J. Control para estabilizar y atenuar las perturbaciones en un péndulo invertido rotatorio. *Rev. Mex. Física E* 2012, *58*, 107–112.
- Han, H.; Zhang, J.; Yang, H.; Hou, Y.; Qiao, J. Data-driven robust optimal control for nonlinear system with uncertain disturbances. *Inf. Sci.* 2023, 621, 248–264. [CrossRef]
- Rumbo-Morales, J.Y.; Ortiz-Torres, G.; Sarmiento-Bustos, E.; Rosales, A.M.; Calixto-Rodriguez, M.; Sorcia-Vázquez, F.D.; Pérez-Vidal, A.F.; Rodríguez-Cerda, J.C. Purification and production of bio-ethanol through the control of a pressure swing adsorption plant. *Energy* 2024, 288, 129853. [CrossRef]
- Fan, Z.X.; Chandra Adhikary, A.; Li, S.; Liu, R. Anti-disturbance inverse optimal control for systems with disturbances. *Optim.* Control Appl. Methods 2023, 44, 1321–1340. [CrossRef]
- Ortiz Torres, G.; Rumbo Morales, J.Y.; Ramos Martinez, M.; Valdez-Martínez, J.S.; Calixto-Rodriguez, M.; Sarmiento-Bustos, E.; Torres Cantero, C.A.; Buenabad-Arias, H.M. Active Fault-Tolerant Control Applied to a Pressure Swing Adsorption Process for the Production of Bio-Hydrogen. *Mathematics* 2023, *11*, 1129. [CrossRef]

- 8. Brizuela-Mendoza, J.A.; Sorcia-Vázquez, F.D.; Rumbo-Morales, J.Y.; Lozoya-Ponce, R.E.; Rodríguez-Cerda, J.C. Active fault tolerant control based on eigenstructure assignment applied to a 3-DOF helicopter. *Asian J. Control* **2021**, *23*, 673–684. [CrossRef]
- You, L.; Yang, X.; Wu, S.; Li, X. Finite-time stabilization for uncertain nonlinear systems with impulsive disturbance via aperiodic intermittent control. *Appl. Math. Comput.* 2023, 443, 127782. [CrossRef]
- 10. Mobayen, S.; Bayat, F.; ud Din, S.; Vu, M.T. Barrier function-based adaptive nonsingular terminal sliding mode control technique for a class of disturbed nonlinear systems. *ISA Trans.* **2023**, *134*, 481–496. [CrossRef]
- Cruz-Rojas, A.; Rumbo-Morales, J.Y.; de la Cruz-Soto, J.; Brizuela-Mendoza, J.A.; Sorcia-Vázquez, F.D.J.; Martínez-García, M. Simulation and control of reactants supply and regulation of air temperature in a pem fuel cells system with capacity of 50 kW. *Rev. Mex. Ing. Quím.* 2019, 18, 349–360.
- Martínez García, M.; Rumbo Morales, J.Y.; Torres, G.O.; Rodríguez Paredes, S.A.; Vázquez Reyes, S.; Sorcia Vázquez, F.d.J.; Pérez Vidal, A.F.; Valdez Martínez, J.S.; Pérez Zúñiga, R.; Renteria Vargas, E.M. Simulation and State Feedback Control of a Pressure Swing Adsorption Process to Produce Hydrogen. *Mathematics* 2022, 10, 1762. [CrossRef]
- Beltrán, A.; Rumbo, J.; Azcaray, H.; Santiago, K.; Calixto, M.; Sarmiento, E. Simulación y control de la velocidad y par electromagnético de un motor de inducción trifásico: Un enfoque a vehículos eléctricos. *Rev. Iberoam. Autom. E Inf. Ind.* 2019, 16, 308–320. [CrossRef]
- 14. Li, J.; Zhang, L.; Li, S.; Mao, Q.; Mao, Y. Active Disturbance Rejection Control for Piezoelectric Smart Structures: A Review. *Machines* **2023**, *11*, 174. [CrossRef]
- Sanchez, R.O.; Rumbo Morales, J.Y.; Ortiz Torres, G.; Pérez Vidal, A.F.; Valdez Resendiz, J.E.; Sorcia Vázquez, F.D.J.; Nava, N.V. Discrete State-Feedback Control Design with D-Stability and Genetic Algorithm for LED Driver Using a Buck Converter. *Int. Trans. Electr. Energy Syst.* 2022, 2022, 8165149. [CrossRef]
- 16. Zhang, X.; Zhang, L.; Zhao, X.; Zhao, N. Reachable set control for singular systems with disturbance via sliding mode control. *J. Frankl. Inst.* **2023**, *360*, 3307–3329. [CrossRef]
- 17. Guha, D.; Roy, P.K.; Banerjee, S. Adaptive fractional-order sliding-mode disturbance observer-based robust theoretical frequency controller applied to hybrid wind–diesel power system. *ISA Trans.* **2023**, *133*, 160–183. [CrossRef]
- Cao, Z.; Niu, B.; Zong, G.; Zhao, X.; Ahmad, A.M. Active disturbance rejection-based event-triggered bipartite consensus control for nonaffine nonlinear multiagent systems. *Int. J. Robust Nonlinear Control* 2023, 33, 7181–7203. [CrossRef]
- Iturriaga-Medina, S.; Martínez-Rodriguez, P.R.; Juarez-Balderas, M.; Sosa, J.M.; Limónes, C.A. A buck converter controller design in an electronic drive for LED lighting applications. In Proceedings of the 2015 IEEE International Autumn Meeting on Power, Electronics and Computing, ROPEC 2015, Ixtapa, Mexico, 4–6 November 2015. [CrossRef]
- 20. Srimathi, R.; Sitoke, S.; Hemamalini, S. High Efficiency Buck LED Driver Using SiC. Energy Procedia 2017, 117, 224–235. [CrossRef]
- Li, G.; Yu, S.; Shi, J. Dynamic response analysis of buck driver for LED based on second-order model. *Electron. Lett.* 2016, 52, 2005–2007. [CrossRef]
- Lin, Y.L.; Kuo, C.C.; Leng, C.M.; Sun, C.C. A High Power Factor LED Driver with Dual Sampling Loop Feedback Control. *IEICE Electron. Express* 2022, 19, 20220178. [CrossRef]
- Chinchero, H.; Alonso, M. Using Magnetic Control of DC-DC Converters in LED Driver Applications. *IEEE Lat. Am. Trans.* 2021, 19, 297–305. [CrossRef]
- 24. Kim, M.G. Discrete time domain modeling and design of current mode controlled flyback LED driver. *Sci. Rep.* **2023**, *13*, 6340. [CrossRef]
- 25. Jiang, W.Z.; Hwu, K.I.; Shieh, J.J. Four-Channel Buck-Type LED Driver with Automatic Current Sharing and Soft Switching. *Appl. Sci.* **2022**, *12*, 5842. [CrossRef]
- Yau, Y.T.; Hwu, K.I.; Tsai, Y.D. Development of Four-Channel Buck-Type LED Driver with Automatic Current Sharing. *Energies* 2021, 14, 7844. [CrossRef]
- Nishikawa, M.; Ishizuka, Y.; Matsuo, H.; Shigematsu, K. An LED drive circuit with constant-output-current control and constant-luminance control. In Proceedings of the INTELEC, International Telecommunications Energy Conference (Proceedings), Providence, RI, USA, 10–14 September 2006. [CrossRef]
- Van Der Broeck, H.; Sauerländer, G.; Wendt, M. Power driver topologies and control schemes for LEDs. In Proceedings of the Conference Proceedings-IEEE Applied Power Electronics Conference and Exposition-APEC, Anaheim, CA, USA, 25 February–1 March 2007; pp. 1319–1325. [CrossRef]
- Garcia, M.; Chatterjee, A.; Ruina, A.; Coleman, M. The simplest walking model: Stability, complexity, and scaling. *J. Biomech. Eng.* 1998, 120, 281–288. [CrossRef]
- 30. Craig, J.J. Robótica; Pearson Prentice Hall: Upper Saddle River, NJ, USA, 2006.
- Di Loreto, M.; Lafay, J.F.; Loiseau, J.J.; Lu, H. On the structure at infinity for linear time-delay systems. *IFAC Proc. Vol.* 2010, 43, 156–161. [CrossRef]
- 32. Pandolfi, L. Disturbance decoupling and invariant subspaces for delay systems. Appl. Math. Optim. 1986, 14, 55–72. [CrossRef]
- 33. Malabre, M.; Rabah, R. Structure at infinity, model matching and disturbance rejection for linear systems with delays. *Kybernetika* **1993**, *29*, 485–498.
- 34. Eves, H.V.; Tonda, J.A.; Magallón, E.T. Funciones de Variable Compleja; Compa. Edit. Continental: Mexico City, Mexico, 1968.

- Figueroa, M.; Bonilla, M.; Malabre, M.; Martinez, J. On failure detection by inversion techniques. In Proceedings of the 2004 43rd IEEE Conference on Decision and Control (CDC) (IEEE Cat. No. 04CH37601), Nassau, Bahamas, 14–17 December 2004; Volume 5, pp. 4770–4775. [CrossRef]
- 36. Estrada, M.B.; Garcia, M.F.; Malabre, M. Time Domain Left Invertibility: Application to Failure Detection. *IFAC Proc. Vol.* 2004, 37, 573–578. [CrossRef]
- 37. Malabre, M.; Kucera, V. Infinite structure and exact model matching problem: A geometric approach. *IEEE Trans. Autom. Control* **1984**, *29*, 266–268. [CrossRef]
- 38. Bonilla, M.; Figueroa, M.; Malabre, M. Solving the Diophantine equation by state space inversion techniques: An illustrative example. In Proceedings of the 2006 American Control Conference, Minneapolis, MN, USA, 14–16 June 2006; p. 6. [CrossRef]
- 39. Willems, J.C. The geometric approach to control system design: A tutorial introduction to a few of the main ideas. In Proceedings of the IEEE Conference on Decision and Control, Athens, Greece, 10–12 December 1986; pp. 1277–1278. [CrossRef]
- 40. Clark, W.; Oprea, M.; Graven, A.J. A Geometric Approach to Optimal Control of Hybrid and Impulsive Systems. *arXiv* 2021, arXiv:2111.11645.
- 41. Commault, C.; Dion, J.M.; Hovelaque, V. A geometric approach for structured systems: Application to disturbance decoupling. *Automatica* **1997**, *33*, 403–409. [CrossRef]
- 42. Marro, G. The Geometric Approachto Control; Italy, 2000.
- 43. Wonham, W.M. Linear Multivariable Control; Springer: Berlin, Germany, 1985. [CrossRef]
- 44. Commault, C.; Dion, J.M. A Comprehensive Introduction to the Geometric Theory of Linear Multivariable Systems. *IEEE Trans. Educ.* **1992**, *35*, 92–97. [CrossRef]
- 45. Commault, C.; Dion, J.M.; Do, T.H. Sensor location for the disturbance rejection by measurement feedback problem. *IFAC Proc. Vol.* **2010**, *43*, 166–171. [CrossRef]
- 46. Basile, G.; Marro, G. Controlled and Conditioned Invariants in Linear System Theory; Prentice Hall: Englewood Cliffs, NJ, USA, 1992.
- 47. Torres, G.O.; Rumbo-Morales, J.Y.; Sánchez, R.O.; Martínez-García, M.; Blanco, M.A.R. Fault Tolerant Control via Input-Output Linearization Method for LED-Driver Using a Boost Converter. *IEEE Access* **2023**, *11*, 10390–10397. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.